**Author’s response to reviews**

**Title:** Prevalence and antimicrobial susceptibility of asymptomatic bacteriuria among women with pelvic organ prolapse in Abakaliki, South-East Nigeria.

**Authors:**
Kenneth Ekwedigwe (ekwedigwe@yahoo.com)
Ileogben Sunday-Adeoye (juladeoye@yahoo.com)
Monday Eliboh (osazemonday@gmail.com)
MARADONA ISIKHUEMEN (maradona4real2002@yahoo.com)
Henry Uro-Chukwu (henryuchukwu@yahoo.co.uk)
Paul Ezeonu (olisaemekapaulo@yahoo.com)
Babafemi Daniyan (babafemidaniyan@yahoo.com)
Emmanuel Yakubu (emayaks@yahoo.com)

**Version:** 3  **Date:** 25 Feb 2018

**Author’s response to reviews:**

1. We note that the order of authors has changed since original submission of the manuscript. In line with COPE guidelines, BioMed Central requires written confirmation from all authors that they agree with any proposed changes in authorship of submitted manuscripts or published articles.

In such cases, we use a standardised form which we would request you and your co-authors to complete. The change in authorship form can be found from the link on the following page:  
[https://www.biomedcentral.com/submissions/editorial-policies#authorship](https://www.biomedcentral.com/submissions/editorial-policies#authorship)

All instructions can be found on the form, please treat the 'current authorship' section as the original authorship. Please return the form within 14 days by email to the Editorial Office with all author signatures (including those newly added/removed). Response :Please use the original author list submitted
2. In your “ethical approval and consent to participate” section of your declarations please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate. Response: This has been included

3. In your “ethical approval and consent to participate” section of your declarations please confirm whether written consent was obtained from participants. If written consent was not obtained, please explain why not. Response: Participants signed consent

4. Please include detailed contributions of each author in the "author contributions" section of your manuscript. Response: This has been included

For example, "FC analyzed and interpreted the patient data regarding the hematological disease and the transplant. RH performed the histological examination of the kidney, and was a major contributor in writing the manuscript. All authors read and approved the final manuscript."

5. We note that the current submission contains some textual overlap with other previously published works, in particular:


This overlap mainly exists in your methods section

and

This overlap mainly exists in your background section.

While we understand that you may wish to express some of the same ideas contained in these publications, please be aware that we cannot condone the use of text from previously published work.

Please re-phrase these sections to minimise overlap. Response: This has been done.

6. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: This has been adhered to. Thanks.